Peer-review report(s). Authors must resolve all issues in the manuscript that are raised in the peer-review report(s) and provide point-by-point responses to each of the issues raised in the peer-review report(s):

Reviewer #1:

Scientific Quality: Grade B (Very good)

Language Quality: Grade B (Minor language polishing)

Conclusion: Accept (General priority)

Specific Comments to Authors:

It is a well-design study adding new information to the literature. Authors in a clear and simply way managed to give their results as well as the relative literature. I have no comments to make and in my opinion the article can be published unaltered.

Revise: Thank you for the reviewers’ comments concerning our manuscript entitled “Jejunal sarcomatoid carcinoma: rare case report and literature review” (Manuscript NO: 95466). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. I am Non-native speakers of English authors, I had provided a language editing certificate by a professional English language editing company which will serve to verify that the language of the manuscript has reached Grade A. I have made correction which we hope meet with approval.